Author's response to reviews

Title: Primary tubercular liver abscess in an immunocompetent adult: a case report

Authors:

CP Baveja (bavejac@indiatimes.com)
Vidhyanidhi Gumma (vidya_micro@rediffmail.com)
Monica Chaudhary (drmonica74@gmail.com)
Himanshu Jha (himanshumamc@gmail.com)

Version: 3 Date: 13 December 2008

Author's response to reviews: see over
To,
The editor,
JMCR.

Subject: reply to the queries of the reviewers

Respected Sir/ madam,
Thank you for considering the case report for your journal. We are thankful to the esteemed reviewers for sparing their time and giving us valuable inputs. Following are the replies for their queries:

Reviewer 1:
1. The abbreviation TLA has been elaborated in the text of abstract.
2. The abscess had decreased to the size 3.1 X 3.8 X 5.1 cm after 6 weeks of therapy and accordingly the liver size also decreased. Unfortunately, the image quality of the repeat USG was very suboptimal as compared to the earlier USG and hence we could not submit it.
3. Apart from the HIV testing (which was non reactive), we did the complete hemogram. There were no other clinical signs and symptoms of any other immunodeficiency syndrome and hence we used the term immunocompetent.
4. As we mentioned earlier, unfortunately this is the only image which we have.

Reviewer 2:
1. the drug sensitivity report has been added in the text(case report section)
2. The identification was done by NAP test and the same has now been added.
3. The procedural details of the PCR have been now elaborated in the text.

I hope this satisfies the esteemed reviewers and looking forward to an early response from you.

Regards
Dr. Monica Chaudhary